Author's response to reviews

Title: Time Esophageal pH<4 Overestimates the Prevalence of Pathologic Esophageal Reflux in Subjects with Gastroesophageal Reflux Disease Treated with Proton Pump Inhibitors

Authors:

Lauren Geerson (lgerson@stanfordmed.org)
George Triadafilopoulos (vagt@standord.edu)
Peyman Sahbaie (psahbaie@standford.edu)
Winston Young (winston@blossomtech.com)
Sheldon Sloan (SSloan@CORUS.JNJ.com)
Malcolm Robinson (mlcolm-robinson@ouhsc.edu)
Philip B. Miner, Jr. (philip-miner@ouhsc.edu)
Jerry D. Gardner (gardnerj@verizon.net)

Version: 2 Date: 9 January 2008

Author's response to reviews:

I have added sections covering competing interests and authors' contributions to the manuscript. I have also corrected a typo in Dr. Gerson's name.